Author’s response to reviews

Title: Tuberculosis screening of travelers to higher-incidence countries: A cost-effectiveness analysis

Authors:

Michael Tan (michael_tan325@yahoo.com)
Dick Menzies (dick.menzies@mcgill.ca)
Kevin Schwartzman (kevin.schwartzman@mcgill.ca)

Version: 6 Date: 2 June 2008

Author’s response to reviews:

p. 12, top paragraph: A sentence break needs to be inserted as follows:

"In fact, the single post-trip tuberculin testing strategy...21.8 months or longer. For Haitian-born travelers from the US returning to Haiti...stays up to 24 months."

Figure 2 has been resubmitted, as the bottom part of the legend for the X-axis was cut off.